Dear editor,

We would like to thank Reviewers for taking the time and effort necessary to review the manuscript. We sincerely appreciate all valuable comments and suggestions, which helped us to improve the quality of the manuscript. Here is a point-by-point response to the reviewers’ comments and concerns.

1. Upon a thorough reading, it appears that the title may not entirely reflect the conclusions derived from the entire manuscript.
   
   **Response:** Thank you for your comments and suggestions that allowed us to greatly improve the quality of the manuscript. We have modified the title according to the reviewer's suggestion: *Prognostic Value of Ultrasound in Early Arterial Complications Post Liver Transplant*.

2. To augment the manuscript's utility for readers, I recommend that the authors elaborate on adding imagings associated with each specific type of complication. These expanded imagings will contribute to a more comprehensive and robust understanding on the topic.
   
   **Response:** According to your suggestion, we have added several typical pictures to help readers better understand the content of the article.

3. Furthermore, I have noted several grammatical and spelling errors throughout the manuscript. To elevate the overall clarity of the content, I would like to recommend thorough language editing.
   
   **Response:** We have sent the revised manuscript to a professional English language editing company for language editing and obtained the relevant language editing certificate.